**Author's response to reviews**

**Title:** Extensive psoriasis induced by Pegylated Interferon: a contraindication to continuing the therapy of HCV-related chronic hepatitis?

**Authors:**

Vincenzo Citro (v.citro@libero.it)
Raffaele Fristachi (r.fristachi@libero.it)
Giovanni Tarantino (tarantin@unina.it)

**Version:** 2  **Date:** 12 January 2007

**Author's response to reviews:**

Dear Editor, We removed the subtitlee, edit the title adding A Case Report, changed the term Summary in Abstract, included legend of figure in the text, included a competing interest declaration, included the Author's contribution, the patient consent, removed two figures and check the manuscript.